Author's response to reviews

Title: The value of maternal anthropometric characteristics in predicting low birth weight and preterm birth in Sudanese newborn infants

Authors:

Eltahir M Elshibly (elshibly@hotmail.com)
Gerd Schmalisch (gerd.schmalisch@charite.de)

Version: 4 Date: 14 May 2008

Author's response to reviews:

Medical Faculty (Charité) of the Humboldt University Berlin

PD Dr. Gerd Schmalisch
Clinic of Neonatology (Charité),
Universitätsmedizin Berlin
Charitéplatz 1
D - 10117 Berlin
Germany
*+49 30 450 516 104
*+49 30 450 516 921
E-Mail: gerd.schmalisch@charite.de

BMC Public Health

Berlin, May 14, 2008

Re: Elshibly EM, Schmalisch G. The effect of maternal anthropometric characteristics and social factors on gestational age and birth weight in Sudanese newborn infants

Dear Dr. Damian Marlee,

Thank you very much for the review of our manuscript and we are grateful for the kind and helpful comments of the referee. It gives me great pleasure to enclose the revised form of the manuscript which has been modified to meet the reviewers’ comments, and to conform with the journal style. I also enclose a detailed protocol of review that delineates all changes to the manuscript.

Thank you,
Sincerely,

Dr. Gerd Schmalisch